

GOPEN ACCESS

Citation: Padmanabhan P, Desikan R, Dixit NM (2020) Targeting TMPRSS2 and Cathepsin B/L together may be synergistic against SARS-CoV-2 infection. PLoS Comput Biol 16(12): e1008461. https://doi.org/10.1371/journal.pcbi.1008461

Editor: Miles P. Davenport, UNSW Australia, AUSTRALIA

Received: May 22, 2020

Accepted: October 22, 2020

Published: December 8, 2020

Copyright: © 2020 Padmanabhan et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its <u>Supporting</u> Information files.

Funding: This work was supported by the DBT/ Wellcome Trust India Alliance Senior Fellowship IA/ S/14/1/501307 to NMD. URL: https://www. indiaalliance.org/. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

RESEARCH ARTICLE

Targeting TMPRSS2 and Cathepsin B/L together may be synergistic against SARS-CoV-2 infection

Pranesh Padmanabhan¹, Rajat Desikan^{2^a}, Narendra M. Dixit^{2,3}*

1 Clem Jones Centre for Ageing Dementia Research, Queesnsland Brain Institute, The University of Queensland, Brisbane, Australia, 2 Department of Chemical Engineering, Indian Institute of Science, Bengaluru, India, 3 Centre for Biosystems Science and Engineering, Indian Institute of Science, Bengaluru, India

¤ Current address: Certara QSP, Certara UK Limited, Sheffield, United Kingdom * narendra@iisc.ac.in

Abstract

The entry of SARS-CoV-2 into target cells requires the activation of its surface spike protein, S, by host proteases. The host serine protease TMPRSS2 and cysteine proteases Cathepsin B/L can activate S, making two independent entry pathways accessible to SARS-CoV-2. Blocking the proteases prevents SARS-CoV-2 entry in vitro. This blockade may be achieved in vivo through 'repurposing' drugs, a potential treatment option for COVID-19 that is now in clinical trials. Here, we found, surprisingly, that drugs targeting the two pathways, although independent, could display strong synergy in blocking virus entry. We predicted this synergy first using a mathematical model of SARS-CoV-2 entry and dynamics in vitro. The model considered the two pathways explicitly, let the entry efficiency through a pathway depend on the corresponding protease expression level, which varied across cells, and let inhibitors compromise the efficiency in a dose-dependent manner. The synergy predicted was novel and arose from effects of the drugs at both the single cell and the cell population levels. Validating our predictions, available in vitro data on SARS-CoV-2 and SARS-CoV entry displayed this synergy. Further, analysing the data using our model, we estimated the relative usage of the two pathways and found it to vary widely across cell lines, suggesting that targeting both pathways in vivo may be important and synergistic given the broad tissue tropism of SARS-CoV-2. Our findings provide insights into SARS-CoV-2 entry into target cells and may help improve the deployability of drug combinations targeting host proteases required for the entry.

Author summary

The COVID-19 pandemic has triggered urgent efforts to repurpose available drugs for the treatment of SARS-CoV-2 infection. Synergistic drug combinations are particularly desirable because they allow the achievement of desired efficacies with minimal toxicities. Here, we predict that drugs targeting the host proteases, TMPRSS2 and Cathepsin B/L, which facilitate SARS-CoV-2 entry into cells through independent pathways, would be

synergistic. We identified the synergy using a mathematical model and found evidence for it in available *in vitro* data. We found the estimated usage of the two pathways to vary vastly across cell types, highlighting the need to target both pathways. Our findings would help maximize the impact of combination therapies targeting host proteases involved in SARS-CoV-2 entry, which are currently in clinical trials.

Introduction

As of August 13, 2020, the severe acute respiratory syndrome coronavirus 2 (SARS-CoV-2) had infected a total of over 20 million people and caused over 744,000 deaths across the world, starting from the first reported infections in Wuhan, China in December, 2019 [1]. SARS-CoV-2 is the third of the major coronavirus outbreaks this century, following SARS coronavirus (SARS-CoV) in 2002-03 and Middle East respiratory syndrome coronavirus (MERS-CoV) in 2012–13 [2]. The scale of the current pandemic far exceeds the earlier two and calls for urgent interventions. At the time of writing, no approved drugs or vaccines existed for any of these coronaviruses [3,4]. Massive efforts are ongoing to 'repurpose' drugs approved for unrelated conditions that may have an effect on SARS-CoV-2 [5-9]. For instance, remdesivir, a drug active against SARS-CoV and MERS-CoV, has been approved for limited use in severe cases of SARS-CoV-2 infection [10]. The steroid dexamethasone has also been approved for use in severe infections [11]. The anti-influenza drug favipiravir has been approved, on the other hand, for use in moderate infections [12]. The anti-malarial drug hydroxychloroquine, which has shown some activity against SARS-CoV-2 in vitro, has been approved in some countries as a prophylactic for high risk groups such as healthcare providers [13,14], although recent studies have questioned its ability to prevent infection of lung cells [15,16]. Several such drugs are under clinical trials, including type I interferons and the HIV-1 drugs lopinavir and ritonavir [17]. Simultaneously, efforts are underway to develop effective vaccines [4,18].

An important class of drugs under investigation putatively targets host factors involved in the lifecycle of SARS-CoV-2, which are being identified through exhaustive virus-host interactome maps [7,19]. Within this class, drugs that block SARS-CoV-2 entry into target cells may be particularly promising, for they would protect cells from becoming infected by the virus and may thus minimize disease. SARS-CoV-2 entry requires the binding of its spike protein, S, to the host cell surface receptor angiotensin-converting enzyme 2 (ACE2) [20,21]. In addition, S must be cleaved by host proteases, as with other coronaviruses [22] and influenza viruses [23], for successful entry, a step often limiting the zoonotic potential of coronaviruses [24]. In vitro studies have suggested that this cleavage, also termed S protein activation, can be accomplished by the transmembrane serine protease TMPRSS2 and endosomal cysteine proteases Cathepsin B and Cathepsin L [21,25]. These proteins have also been implicated in the entry of SARS-CoV [26-30], to which SARS-CoV-2 is closely related [31], and have formed the basis for corresponding studies on SARS-CoV-2. (MERS-CoV uses a different receptor, dipeptidyl peptidase 4, but appears to use the same proteases for activation [32,33].) ACE2, TMPRSS2 and Cathepsin B/L, thus, are key host targets for blocking SARS-CoV-2 entry. ACE2 plays a critical role in regulating several vital parameters, such as blood pressure, and targeting it could come with serious risks [34]. Targeting host proteases appears to have fewer side effects. Indeed, camostat mesylate, a drug that targets TMPRSS2, has been approved for use in the treatment of chronic pancreatitis and postoperative reflux esophagitis in Japan [35]. A human clinical trial has now been initiated to assess the possibility of using camostat mesylate monotherapy for treating SARS-CoV-2 infection (Clinical-Trials.gov ID: NCT04321096). Nafamostat mesylate, another drug targeting TMPRSS2 and

approved in Japan as well as by the FDA, blocked SARS-CoV-2 entry more potently than camostat mesylate *in vitro* [36] and is also under clinical trials (ClinicalTrials.gov ID: NCT04352400). Drugs targeting Cathepsin B/L are under development [37,38]. Hydroxychloroquine is thought to act against SARS-CoV-2 via several mechanisms, including blocking the endocytic pathway; it suppresses clathrin-mediated virus uptake and also prevents endosomal acidification, the latter required for Cathepsin B/L activity [3,16,39]. While studies suggest that hydroxychloroquine monotherapy may be inadequately efficacious [15,16], its efficacy in combination with camostat mesylate is under clinical evaluation (ClinicalTrials.gov ID: NCT04338906).

The proteases TMPRSS2 and Cathepsin B/L are unrelated and work independently [22], suggesting that SARS-CoV-2 can enter cells via two independent pathways. Studies on SARS--CoV have shown that TMPRSS2 is expressed on the target cell surface and acts by cleaving S and facilitating the fusion of viral and cell membranes at the target cell surface, whereas Cathepsin B/L are expressed in endosomes and act after the virus has been endocytosed, facilitating the fusion of viral and endosomal membranes [26–30]. It appears, thus, that blocking either TMPRSS2 or Cathepsin B/L may be insufficient to block virus entry; the virus could continue to enter cells via the pathway that remains unblocked. Indeed, *in vitro* studies have shown, both with SARS-CoV and SARS-CoV-2, that simultaneous targeting of both TMPRSS2 and Cathepsin B/L is required for fully stopping virus entry [21,28]. The pathways have not been studied in SARS-CoV-2 infection *in vivo*.

Given the clinical trials underway with drugs blocking SARS-CoV-2 entry, we were interested in assessing whether combination therapy that simultaneously targeted both the entry pathways would be advantageous over therapies targeting the pathways individually. To this end, we developed a mathematical model of SARS-CoV-2 dynamics under treatment with a TMPRSS2 inhibitor, a Cathepsin B/L (or endosomal pathway) inhibitor, or both. We explicitly accounted for the dependence of viral entry efficiency via the two pathways on the expression levels of the two proteases and the effects of the drugs in suppressing the respective efficiencies in a dose-dependent manner. Our model predicted that targeting the two pathways simultaneously was likely not only to be efficacious but also synergistic.

Synergy between drugs implies that their combined effect is more than their individual effects added together [40-43]. Synergistic drug combinations are preferable because they allow the realization of the desired efficacy with lower net drug exposure, thus reducing toxicity. Conversely, at dosages limited by toxicities, synergistic drug combinations achieve higher efficacies than their non-synergistic counterparts. Synergy is thought to arise typically as the result of downstream interactions between the steps/pathways targeted by the drugs [40-43]. The synergy we predict here is thus surprising because the two entry pathways targeted are independent. Using our modelling and analysis, we show how this synergy arises from effects both at the single cell level as well as at the cell population level. We show further that the synergy is evident, although unrecognized, in available *in vitro* experiments with SARS-CoV-2 and SARS-CoV, providing evidence in strong support of our model predictions.

While the availability of the two pathways for entry is well recognized, their relative usage has not been established except in cell types where one of the pathways dominates overwhelmingly [21,28]. Our formalism allowed us to analyse available *in vitro* data and quantify the relative usage of the two pathways in different cell types. We estimated thus that in Vero cells expressing TMPRSS2, infection proceeded ~65% of the time via the TMPRSS2 pathway and ~35% via the Cathepsin B/L pathway. The usage was similar with SARS-CoV. The usage, however varied widely across cell lines, suggesting, given the potentially broad tissue tropism of SARS-CoV-2, extending beyond the lungs [21,44], that targeting both pathways may be important and synergistic *in vivo*. Together, our findings could help optimize combination therapies targeting TMPRSS2 and Cathepsin B/L for preventing SARS-CoV-2 entry.

Results

Overview of the mathematical model

We considered *in vitro* experiments where a population of target cells is exposed in the presence or absence of protease inhibitors to virions expressing the SARS-CoV-2 spike protein, S, and the extent of infection is measured. We constructed a mathematical model of the ensuing dynamical processes, focusing on virus entry and the role of the protease inhibitors (Fig 1). We assumed that cells could get infected following S protein activation by either TMPRSS2 or Cathepsin B/L. The efficiency of virus entry via the two pathways depended on the expression levels of the respective proteases. We denoted as S_t the susceptibility of a target cell expressing n_t copies of TMPRSS2 to entry via the TMPRSS2 pathway. The susceptibility here was also the probability with which a virion could successfully enter the cell when TMPRSS2 was the sole limiting factor. Analogously, S_c was the susceptibility to entry via the Cathepsin B/L pathway of a cell expressing n_c copies of Cathepsin B/L. The overall susceptibility, which we denoted S_{tc} . of a cell expressing n_t and n_c copies of the two proteases, respectively, was then $S_{tc} = S_t + S_c - S_t S_{cc}$ indicating the independence of the two pathways. We let S_t and S_c increase with n_t and n_{α} respectively, following distinct Hill functions (Methods). We assumed that drugs acted by suppressing the proteases in a dose-dependent manner. A TMPRSS2 inhibitor thus blocked a fraction of the n_t TMPRSS2 molecules, reducing S_t . A Cathepsin B/L inhibitor similarly lowered S_c Furthermore, we assumed that a TMPRSS2 inhibitor left S_c unaffected, so that entry could proceed uninhibited via the Cathepsin B/L pathway even when a TMPRSS2 inhibitor was present in excess. Similarly, a Cathepsin B/L inhibitor would let entry occur uninhibited via the TMPRSS2 pathway. The drugs too thus acted independently. Only when both the drugs were used could a cell be fully protected (Fig 1A). We recognized that the cells in culture exhibited a distribution of the expression levels of TMPRSS2 and Cathepsin B/L and thus had a distribution of susceptibilities to infection via the two pathways (Fig 1B). The effects of the drugs too, thus, exhibited cell-to-cell variability. With this description, we formulated dynamical equations to predict how the population of cells in culture would get infected and how the drugs would suppress the infection (Methods). The model was solved using parameter values listed in <u>S1 Table</u>. Based on the model, we derived analytical expressions of the synergy between the drugs and elucidated its origins (S1 Text and S2 Text). Finally, we applied the model to analyse in vitro data and estimated the relative usage of the two pathways (Methods).

The activity of TMPRSS2 and Cathepsin B/L inhibitors depends on the relative usage of the respective pathways

We considered cells expressing TMPRSS2 and Cathepsin B/L following independent, log-normal distributions (Fig 2A and 2B). The susceptibilities, S_t and S_c , of the cells to entry via the two pathways were modelled as rising sigmoidally from zero at low expression levels to 1 at high expression levels of the respective proteases (Fig 2A and 2B). The joint distribution of the expression levels of TMPRSS2 and Cathepsin B/L on cells was given by the product of the individual distributions (Fig 2C). The overall susceptibility, S_{tc} , was predicted to be small when both proteases were at low expression levels, rose with the expression levels, and reached 1 when either expression level was high (Fig 2D).

With this description of the susceptibility of cells to entry, we examined first the influence of TMPRSS2 expression levels on virus entry. We considered scenarios where the mean TMPRSS2 level was low, medium, or high, which corresponded to S_t at the mean expression levels of 0.001, 0.5, and 0.95, respectively (Fig 2E). Note that TMPRSS2 levels can vary substantially across tissues in the body, with the prostate, for instance, displaying significantly higher



Fig 1. Schematic of the mathematical model incorporating the two independent pathways of SARS-CoV-2 entry into target cells. (A) After the SARS-CoV-2 virion attaches to the host surface receptor ACE2, S protein activation occurs by the host proteases TMPRSS2 (transmembrane) or Cathepsin B/L (endosomal) thus yielding two independent entry pathways. Blocking both pathways is essential for preventing infection of the cell expressing both proteases. (B) TMPRSS2 and Cathepsin B/L expression levels vary across cells. The entry efficiency through a pathway increases and the efficacy of the inhibitor at a given concentration decreases with the expression level of the corresponding protease.

expression levels than the lung, trachea, or salivary glands [45–48]. Different cell lines too express widely varying TMPRSS2 levels [49]. For the distributions chosen, our model predicted that the total population of cells infected increased with TMPRSS2 expression, going from ~1640 cells/ml at low to ~5050 cells/ml infected at high mean TMPRSS2 expression level (out of 10^5 initial target cells/ml) at 1 d post-infection (Fig 2F). The initial distribution of Cathepsin B/L expression (Fig 2B) was unaltered in these calculations. The susceptibility, S_c , through the Cathepsin B/L pathway at the mean Cathepsin B/L expression level was 0.029. Thus, the utilization of the TMPRSS2 pathway (S_t/S_c) was, on average, ~30-fold lower, 17.1-fold higher, and 32.5-fold higher than the Cathepsin B/L pathway for the low, medium and high TMPRSS2 scenarios, respectively, indicating that as the TMPRSS2 level increased, it became increasingly preferred as the entry pathway in our predictions.

We reasoned that as the preference for the TMPRSS2 pathway increased, it would compromise the activity of a Cathepsin B/L inhibitor. We examined this by performing the calculations above but in the presence of a Cathepsin B/L inhibitor. Mimicking experiments [21], we calculated the fraction of infection events 'unaffected' by the Cathepsin B/L inhibitor, which we denoted $f^{\mu}(D_C)$ (Fig 2G). Thus, $f^{\mu}(D_C) = 1$ would imply that the drug prevented no entry



Fig 2. Predictions of SARS-CoV-2 entry into cells with heterogenous protease expression levels. (A, B) The log-normal distribution of TMPRSS2 (A) and Cathepsin B/L (B) across cells. Dependence of the susceptibility of infection through the TMPRSS2 pathway on the TMPRSS2 expression level, n_t , (A) and through the Cathepsin B/L pathway on the Cathepsin B/L expression level, n_c (B). (C) The joint distribution of cells expressing both TMPRSS2 and Cathepsin B/L. (D) Dependence of susceptibility of infection on both TMPRSS2 and Cathepsin B/L expression levels. In (A)-(D), $\overline{n_t} = 11.5$ and $\overline{n_c} = 11.5$. (E) The log-normal distribution of TMPRSS2 across cells with low ($\overline{n_t} = 10$), medium ($\overline{n_t} = 11.7$) and high ($\overline{n_t} = 12.4$) mean TMPRSS2 expression levels. (F, G) Infected cells in the absence of inhibitors (F) and fraction of infection events uninhibited by different concentrations of a Cathepsin B/L across cells with low ($\overline{n_c} = 10.7$), medium ($\overline{n_c} = 12.4$) and high ($\overline{n_c} = 13.1$) mean Cathepsin B/L expression levels. (I, J) Infected cells in the absence of inhibitors (I) and fraction of infection events uninhibited by different concentrations of TMPRSS2 expression ($\overline{n_c} = 10.7$), medium ($\overline{n_c} = 12.4$) and high ($\overline{n_c} = 13.1$) mean Cathepsin B/L expression levels. (I, J) Infected cells in the absence of inhibitors (I) and fraction of infection events uninhibited by different concentrations of TMPRSS2 expression ($\overline{n_c} = 10.7$), medium ($\overline{n_c} = 12.4$) and high ($\overline{n_c} = 13.1$) mean Cathepsin B/L expression levels. (I, J) Infected cells in the absence of inhibitors (I) and fraction of infection events uninhibited by different concentrations of TMPRSS2 expression ($\overline{n_c} = 11.5$). In (F), (G), (I), and (J), the cumulative level of infection was computed at day 1 post infection, and the drug concentrations were normalised by the respective γ values (Methods). The other parameters and initial conditions are listed in S1 Table.

events, whereas $f^{\mu}(D_C) = 0$ would imply that the drug protected all cells that would have been infected otherwise. Our model predicted that as the inhibitor level increased, $f^{\mu}(D_C)$ decreased, indicating increased drug activity (Fig 2G). $f^{\mu}(D_C)$, however, did not reach 0 even at arbitrarily high drug concentrations. Instead, it plateaued at values that depended on the mean TMPRSS2 expression level. When the latter expression level was low, $f^{\mu}(D_C)$ plateaued at ~0.23, indicating strong protection, whereas when the expression level was high, $f^{\mu}(D_C)$ was ~0.93, implying little protection. Thus, the 'apparent' efficacy of the Cathepsin B/L inhibitor was dependent on TMPRSS2 expression. The inhibitor was efficacious when the predominant pathway was via Cathepsin B/L, which here happened when TMPRSS2 expression was low. When the latter expression was high, Cathepsin B/L was utilized negligibly; blocking it would thus have a minimal effect on virus entry, rendering the inhibitor poorly efficacious.

In the same way, we predicted that the apparent activity of TMPRSS2 inhibitors would depend on the expression level of Cathepsin B/L. When we used low, medium or high mean expression levels of Cathepsin B/L in our calculations (Fig 2H), with the TMPRSS2 expression level unaltered (Fig 2A), we found that the extent of infection increased in the absence of drugs from ~2960 cells/ml at low to ~5390 cells/ml at high Cathepsin B/L expression (Fig 2I). With a TMPRSS2 inhibitor, the extent of infection decreased, but the fraction unaffected, now denoted $f^{\mu}(D_T)$, plateaued at values that increased with the mean Cathepsin B/L expression level– $f^{\mu}(D_T)$ ~0.13 at low and $f^{\mu}(D_T)$ ~0.88 at high expression levels–indicating again that the apparent activity of the TMPRSS2 inhibitor decreased as the Cathepsin B/L pathway became increasingly preferred (Fig 2J).

These predictions were consistent with several experiments showing that high expression levels of TMPRSS2 led to poor efficacies of Cathepsin B/L inhibitors [21,27–29]. For instance, when 293T cells expressing human ACE2 but not TMPRSS2 were infected with SARS-CoV in the presence of 25 mM NH₄Cl, which prevents endosomal acidification and hence blocks Cathepsin B/L, the extent of infection reduced ~40-fold over that in the absence of NH₄Cl [29]. When the same cells were engineered to express TMPRSS2, the reduction was just ~4-fold, marking a drastic loss of the apparent activity of NH₄Cl. Similar effects were observed with other endosomal pathway inhibitors including E-64d, MDL28170, EST, and bafilomycin [21,27–29]. Importantly, recent experiments observed this effect with SARS-CoV-2 [21]. When 293T cells expressing ACE2 were infected with SARS-CoV-2 pseudotyped viruses, treatment with E-64d reduced the infection by 90%, whereas when the cells additionally expressed TMPRSS2, the effect of E-64d nearly vanished [21].

The predictions were also consistent with experiments showing high activity of TMPRSS2 inhibitors when the expression level of TMPRSS2 was high [21,28]. For instance, with HeLa cells expressing ACE2, the effect of camostat mesylate was negligible in inhibiting SARS-CoV infection compared to its effect on the same cells engineered to overexpress TMPRSS2 [28]. With the latter cells, camostat mesylate prevented ~65% of infections at the highest doses, consistent with the plateau predicted by our model (Fig 2G and 2J). The implication was that the remaining infection occurred through the Cathepsin B/L pathway. In the same way, with Vero cells, which hardly express TMPRSS2, camostat mesylate had little effect in reducing SARS-CoV-2 infections, whereas E-64d, which blocked Cathepsin B/L reduced nearly 100% of the infections [21]. At the same time, in Vero cells engineered to overexpress TMPRSS2, with the same dosages of the drugs, E-64d could only block ~25% of the infections, whereas camostat mesylate blocked nearly ~65% of the infections, indicative again of the relative utilization of the pathways in these cell lines. Similar results were observed also with SARS-CoV infection [21].

These findings suggest caution when interpreting assays measuring the efficacies of drugs targeting host proteases. A drug may block its target protease potently but may still appear poorly efficacious at preventing viral entry because of the activity of the other protease. More

importantly, the findings suggest that with cells expressing significant levels of both TMPRSS2 and Cathepsin B/L, the use of either a TMPRSS2 inhibitor or a Cathepsin B/L inhibitor alone would be insufficient to block entry completely. The inhibitors would have to be used together to completely block entry. We therefore examined next our model predictions when the inhibitors were used together.

The combined use of TMPRSS2 and Cathepsin B/L inhibitors can fully block entry

To elucidate the effect of combination therapy, we predicted the effects of the drugs alone and in combination in blocking virus entry. Because the cells expressed both proteases, the drugs individually could not block entry fully. For the parameters chosen, the effect of the Cathepsin B/L inhibitor plateaued at $f^{\mu}(D_C) \sim 0.78$ and that of the TMPRSS2 inhibitor at $f^{\mu}(D_T) \sim 0.39$ when used alone (Fig 3A). When the two were combined, our model predicted 100% block of entry; the fraction unaffected, now denoted $f^{\mu}(D_T, D_C)$, plateaued at 0 (Fig 3A).

This prediction was consistent, again, with several *in vitro* studies showing that full blockade of entry required the combined use of TMPRSS2 and Cathepsin B/L inhibitors [21,28]. With HeLa cells expressing ACE2 and TMPRSS2, EST alone blocked ~34% of SARS-CoV infection, camostat mesylate blocked ~58%, and the two together blocked ~100% of the infections [28]. In Vero cells expressing TMPRSS2, the corresponding numbers were ~20% for E-64d, ~47% for camostat mesylate, and ~98% for both with SARS-CoV infection [21]. Importantly, the latter numbers were ~21%, ~57%, and ~92%, respectively, for SARS-CoV-2 infection [21].

TMPRSS2 and Cathepsin B/L inhibitors synergize in blocking virus entry

The above predictions indicated strong synergy between drugs blocking the two entry pathways (Fig 3A). Because the pathways are independent, the combined effect of the drugs is expected to follow Bliss independence, where the fraction of events unaffected by the combination is the product of the fractions unaffected when the drugs are used alone [40,43,50]. In other words, Bliss independence would imply $f_{Bliss}^u(D_T, D_C) = f^u(D_T) \times f^u(D_C)$. If $f^u(D_T, D_C)$ is smaller than $f_{Bliss}^u(D_T, D_C)$, the drugs exhibit synergy; *i.e.*, they block more infections together than expected from their individual effects put together. With the above parameter values, where $f^u(D_T)\sim 0.39$ and $f^u(D_C)\sim 0.78$, Bliss independence would yield $f_{Bliss}^u(D_T, D_C) \sim 0.3$. The effect predicted by our model, however, was $f^u(D_T, D_C) = 0$, implying strong synergy (Fig 3A).

The synergy can be visualized in two ways [40]: horizontal and vertical synergy (Fig 3A). Horizontal synergy refers to the decrease in drug levels from that assuming independent action that is required to produce the same effect as that achieved by Bliss independence. Vertical synergy refers to the gain in efficacy at fixed drug levels over that assuming Bliss independence. The two are equivalent ways of quantifying synergy. The latter has been termed Bliss synergy and is more widely used [43,50–52]. Here, we employed the latter measure throughout and denoted it as $\beta_{Bliss} = f_{Bliss}^u(D_T, D_C) - f^u(D_T, D_C)$. As the extent of synergy increases, β_{Bliss} increases. Note that $0 \le \beta_{Bliss} \le 1$ here. (Negative values of β_{Bliss} indicate antagonistic interactions between the drugs.)

To assess the extent of synergy that could be realized, we performed the above calculations over wide ranges of drug levels. We predicted that the combined efficacy expected assuming Bliss independence was small $(f_{Bliss}^u(D_T, D_C) \sim 1)$ when the drug levels were small and rose to the maximum estimated above $(f_{Bliss}^u(D_T, D_C) \sim 0.3)$ at saturating drug levels (Fig 3B). In striking contrast, the predicted efficacy of the combination was much higher at the saturating drug levels ($f^u(D_T, D_C) \sim 0$) (Fig 3C). Correspondingly, $\beta_{Bliss} \sim 0.3$, quantifying the maximum extent of synergy realized for the parameters chosen (Fig 3D). These predictions were robust to parameter variations (S1 and S2 Figs).



Fig 3. Predictions of the effect of combination treatment targeting both TMPRSS2 and Cathepsin B/L pathways. (A) The fraction of infection events unaffected by Cathepsin B/L inhibitor, TMPRSS2 inhibitor or both for different drug concentrations. The extent of predicted horizontal and vertical or Bliss synergy (see text) are marked. (B, C) The expected effect of the combination from Bliss independence (B) and predicted combination effect (C) over varying drug concentrations. (D) The predicted Bliss synergy over a range of drug concentrations. The parameters and initial conditions are listed in S1 Table.

The predictions were intriguing because the pathways are independent, and the combination should have accordingly followed Bliss independence. We examined next the possible origins of the synergy in our model.

Origin of synergy at the single cell level

Synergy could arise from the effects of the drugs at the single cell level [51,53,54] or at the cell population level [53,55–57]. We sought to examine first whether synergy could arise in our calculations from effects at the single cell level. For this, we considered cells that had the same TMPRSS2 and Cathepsin B/L levels, eliminating heterogeneity in the cell population. Any synergy would then have to arise from effects at the single cell level. We repeated our calculations above (Figs 2 and 3) with this homogeneous cell population.

We found that for fixed Cathepsin B/L levels, as the expression level of TMPRSS2 increased from low to high, the number of cells infected increased from ~870/ml to ~6240/ml in the absence of drugs (Fig 4A). With a Cathepsin B/L inhibitor, $f^{\mu}(D_C)$ decreased and reached a plateau (Fig 4B). The plateau was lower, indicating greater drug efficacy, as the TMPRSS2 level decreased, consistent with the observations above of the drug efficacy increasing as the relative usage of the corresponding pathway increased. Analogous predictions resulted upon increasing Cathepsin B/L expression levels at a fixed TMPRSS2 level (Fig 4C) and in the presence of increasing concentrations of a TMPRSS2 inhibitor (Fig 4D).

For our calculations with both the drugs present simultaneously, we chose protease expression levels such that the individual drug effects plateaued at $f^{\mu}(D_T) \sim 0.32$ and $f^{\mu}(D_C) \sim 0.79$, respectively (Fig 4E). In the absence of synergy, the expected combined effect would plateau at $f^{\mu}_{Bliss}(D_T, D_C) \sim 0.26$. Our model predicted instead that the combined effect plateaued at $f^{\mu}(D_T, D_C) \sim 0.26$. Our model predicted instead that the combined effect plateaued at $f^{\mu}(D_T, D_C) \sim 0.26$. Our model predicted instead that the combined effect plateaued at $f^{\mu}(D_T, D_C) \sim 0.46$. Our model predicted instead that the combined effect plateaued at $f^{\mu}(D_T, D_C) \sim 0.26$. Our model predicted instead that the combined effect plateaued at $f^{\mu}(D_T, D_C) \sim 0.26$. Specifically, the extent of Bliss synergy, denoted β^{cell}_{Bliss} to emphasize its origin, was $\beta^{cell}_{Bliss} \sim 0.26$. Spanning a wide range of drug concentrations, we found that $f^{\mu}_{Bliss}(D_T, D_C)$ decreased from ~1 at low drug levels and plateaued at ~0.26 at high drug levels (Fig 4F), whereas $f^{\mu}(D_T, D_C)$ decreased further and reached zero at high levels (Fig 4G). The corresponding synergy was also low at low drug levels and rose to $\beta^{cell}_{Bliss} \sim 0.26$ at high drug levels, reiterating the existence of synergy at the single cell level and quantifying its extent (Fig 4H).

The above calculations were for fixed TMPRSS2 and Cathepsin B/L levels. To ascertain the robustness of the results, we repeated the calculations with different levels of either TMPRSS2 (Fig 4I) or Cathepsin B/L (Fig 4J), while keeping all the other parameters fixed. We found that the synergy had a non-monotonic dependence on the expression levels. As the TMPRSS2 level increased, for instance, β_{Bliss}^{cell} rose from zero, attained a peak that was dependent on the drug level, and then declined again to zero (Fig 4I). We explain this effect by considering again the relative preferences for the two entry pathways. When the TMPRSS2 level was low, its contribution to entry was low, so that only Cathepsin B/L inhibitors would display significant efficacy. The effect of the TMPRSS2 inhibitor was imperceptible. No synergy could arise. Conversely, when the TMPRSS2 levels were high, Cathepsin B/L inhibitors had hardly any effect. Synergy was again not possible. At intermediate expression levels, synergy could occur and was maximized. Thus, synergy was high when the relative usage of the two pathways were comparable and vanished when one of the two pathways dominated. With increasing drug levels, this effect was amplified because the drugs blocked their respective pathways more effectively, leading to increased overall synergy at intermediate expression levels. Analogous effects were evident with varying Cathepsin B/L expression levels (Fig 4J).

These trends also emerged in analytical expressions we derived for synergy in single round infection assays [21,28], providing a deeper understanding of the synergy at the single cell level (S1 Text). Indeed, predictions using the analytical expressions were in close agreement with those of our model above (S3 Fig).

We examined next whether synergy could also arise from effects at the cell population level.



Fig 4. Predictions of SARS-CoV-2 entry into cells with homogenous protease expression levels. (A, B) Cells infected in the absence of inhibitors (A) and fraction of infection events uninhibited by different concentrations of Cathepsin B/L inhibitor (B) in a population of cells expressing low (n_t = 20,000 copies/cell), medium (n_t = 100,000 copies/cell) and high (n_t = 500,000 copies/cell) TMPRSS2 levels and fixed (n_c = 150,000 copies/cell) Cathepsin B/L level. (**C**, **D**) Cells infected in the absence of inhibitors (C) and fraction of infection events uninhibited by different concentrations of TMPRSS2 inhibitor (D) in a population of cells expressing low (n_c = 20,000 copies/cell), medium (n_c = 170,000 copies/cell) and high (n_c = 500,000 copies/cell) Cathepsin B/L levels and fixed (n_t = 100,000 copies/cell) TMPRSS2 level. (**E**) The fraction of infection events unaffected by Cathepsin B/L inhibitor, TMPRSS2 inhibitor or both for different drug concentrations in cell population with n_t = 100,000 copies/cell. (**F**, **G**) The expected combination effect from Bliss independence (F) and predicted combination effect (G) over a varying levels of drug concentrations. (**H**) The predicted Bliss synergy over a range of drug concentrations. (**I**, **J**) The predicted Bliss synergy for varying TMPRSS2 expression and fixed (n_c = 150,000 copies/cell) Cathepsin B/L expression and fixed (n_t = 100,000 copies/cell) TMPRSS2 expression and fixed (n_c = 150,000 copies/cell) Cathepsin B/L expression and fixed (n_c = 150,000 copies/cell) TMPRSS2 expression and fixed (n_c = 150,000 copies/cell) TMPRSS2 expression and fixed (n_c = 150,000 copies/cell) TMPRSS2 expression and fixed (n_c = 150,000 copies/cell) TMPRSS2 expression and fixed (n_c = 150,000 copies/cell) Cathepsin B/L expression and fixed (n_c = 150,000 copies/cell) TMPRSS2 expression and fixed (n_c = 150,000 copies/cell) TMPRSS2 expression at two different drug concentrations. The drug concentrations were normalised by the respective

Origin of synergy at the cell population level

Synergy at the cell population level can arise when heterogeneity exists in the response of cells to the drugs involved [53,55–57]. Here, such heterogeneity could arise from the distributions of the expression levels of the two proteases across cells. The synergy emerging from such heterogeneity is evident in the extreme scenario we conceived where cells in culture either expressed TMPRSS2 or Cathepsin B/L but not both and for which we were able to derive analytical expressions of the synergy (S2 Text). To assess the extent of synergy arising from the cell population level in a more realistic scenario, we performed calculations akin to those in Figs 2 and 3 above (Methods). We predicted that when the overall synergy, β_{Bliss} , reached a plateau at ~0.3, the cell population level synergy, β_{Rliss}^{popn} , was ~0.13, amounting to a contribution of nearly 43% to the total synergy (Fig 5A). By varying drug levels over wide ranges, we predicted that the synergy would be low at low drug levels, and increase monotonically as drug levels rise (Fig 5B). The single cell level synergy displayed similar trends, leading to the overall synergy also rising monotonically with drug levels and eventually saturating at ~0.3 (Fig 5C). We recognized from our analysis of the extreme scenario above that $\beta_{Bliss}^{popn} \leq 1/4$ (S2 Text), whereas β_{Bliss}^{cell} can rise to 1 (S1 Text). Thus, depending on the overall synergy, the contribution from the cell population level may vary. Whereas it was ~43% above, it could be much lower if β_{Bliss}^{popn} were smaller and/or β_{Bliss}^{cell} were larger. To illustrate this, we performed calculations with parameter settings that yielded a much higher overall synergy as well as a lower cell population level synergy. Here, β_{Bliss} saturated at 0.69, whereas $\beta_{\text{Rliss}}^{\text{popn}}$ saturated at 0.03, resulting in a far smaller contribution, ~4.4%, from the latter (Fig 5D, 5E and 5F). Finally, we note that β_{Bliss}^{popn} too displayed non-monotonic dependencies on the expression levels of the proteases (Fig 5G and 5H). These trends mimic those with the single cell level synergy (Fig 4I and 4J) and are understood from the analytical expressions we derived (S2 Text).

In summary, it followed from these model predictions that contributions could arise to the synergy from effects of the drugs at the single cell as well as cell population levels. We examined next whether the synergy we predicted was evident in available *in vitro* data on SARS-CoV-2 and SARS-CoV infection.

TMPRSS2 and Cathepsin B/L inhibitors display synergy in vitro

We considered data from 4 independent experiments, 1 with SARS-CoV-2 and 3 with SARS-CoV, performed across 2 different cell lines, using camostat mesylate as the TMPRSS2 inhibitor and 3 different drugs as Cathepsin B/L inhibitors, in two independent studies that have examined the combined effect of blocking TMPRSS2 and Cathepsin B/L [21,28]. The studies exposed cells to pseudotyped viruses in the absence of drugs or in the presence of either or both drugs and reported the fraction of infection events unaffected by the drugs, namely, $f^{\mu}(D_T), f^{\mu}(D_C)$, and $f^{\mu}(D_T, D_C)$. Using the latter data, we estimated the extent of Bliss synergy, $\beta_{Bliss} = f^{\mu}(D_T) \times f^{\mu}(D_C) - f^{\mu}(D_T, D_C)$, between the drugs (Methods). We found that these experiments displayed synergy between TMPRSS2 and Cathepsin B/L inhibitors, in agreement with our model predictions (Table 1).

With SARS-CoV-2, we estimated β_{Bliss} ~0.26 between camostat mesylate and E-64d in Vero cells expressing TMPRSS2, suggesting that the combination protected an additional 26% of target cells from being infected relative to the additive effects of the two drugs at the same level of exposure. Strong synergy was also seen with SARS-CoV, with β_{Bliss} ~0.4 under the same conditions. The synergy extended to other drug combinations and cell lines. With camostat mesylate and EST, the latter a broad inhibitor of cysteine proteases including Cathepsin B/L,



Fig 5. Predictions of population level synergy. (**A**-**F**) The predicted population level and overall synergy over a range of drug concentrations (see text for details). (**G**, **H**) The predicted overall and population level synergy for varying mean TMPRSS2 expression and fixed mean ($\overline{n_c} = 11.5$) Cathepsin B/L expression (G) and for varying mean Cathepsin B/L expression and fixed mean ($\overline{n_c} = 11.5$) TMPRSS2 expression at two different drug concentrations (H). The drug concentrations were normalised by the respective γ values. In A-C and G-H, $n_t^{50} = 1.2 \times 10^5$ copies/cell and $n_c^{50} = 2.4 \times 10^5$ copies/cell. In D-F, $n_t^{50} = 4 \times 10^4$ copies/ cell, $n_c^{50} = 4 \times 10^4$ copies/ cell. In G-H, $D_T/\gamma_T = D_C/\gamma_C = 10$. The parameters are listed in S1 Table.

https://doi.org/10.1371/journal.pcbi.1008461.g005

 β_{Bliss} ~0.22 with SARS-CoV infection in HeLa cells expressing ACE2 and TMPRSS2. With camostat mesylate and bafilomycin, which prevents endosomal acidification and thus blocks Cathepsin B/L function, the corresponding estimate was β_{Bliss} ~0.17.

Table 1. Experimental evidence of Bliss synergy. We computed the extent of synergy from reported data on the effects of the individual drugs and the combination. The difference between the expected effect of the combination in terms of the fraction unaffected (f^{μ}) based on Bliss independence and the effect observed yielded the extent of synergy. Infection was by viruses pseudotyped with either SARS-CoV-2 S (denoted SARS-2-S) or SARS-CoV S (denoted SARS-S) protein. The experimental observations ($f^{\mu}(D_T)$, $f^{\mu}(D_C)$, and $f^{\mu}(D_T, D_C)$) reported are reproduced for convenience and using which we computed the extent of synergy (β_{Bliss}).

Infection	Cell line	Drug combination	$f^{\mu}(D_T)$	$f^{\mu}(D_C)$	$f^{\mu}(D_T,D_C)$		β_{Bliss}	Ref.
					Expected	Observed		
SARS-2-S	Vero-TMPRSS2	Camostat (D_T) + E-64d (D_C)	0.43	0.79	0.34	0.08	0.26	[21]
SARS-S	Vero-TMPRSS2	Camostat (D_T) + E-64d (D_C)	0.53	0.8	0.42	0.02	0.4	[21]
SARS-S	HeLa-ACE2-TMPRSS2	Camostat (D_T) + EST (D_C)	0.42	0.66	0.28	0.06	0.22	[28]
SARS-S	HeLa-ACE2-TMPRSS2	Camostat (D_T) + Bafilomycin (D_C)	0.42	0.6	0.25	0.08	0.17	[28]

*Note that synergy is not expected in cell lines such as Calu-3 and Caco-2 where one pathway dominates (see Fig 6).

https://doi.org/10.1371/journal.pcbi.1008461.t001

This evidence of the strong presence of synergy between TMPRSS2 and Cathepsin B/L inhibitors validated our model predictions. The synergy could be exploited in defining optimal drug combinations targeting host proteases required for SARS-CoV-2 entry, as has been suggested in other contexts [40,43,51,52,54]. For such optimization, knowledge of the nature of synergy, *i.e.*, whether it originates from the single cell or the cell population level, is important. We considered this next.

The relative usage of the entry pathways varies widely across cell types

From our calculations above, synergy at the single cell level would depend on the susceptibility of individual cells to entry via both pathways, whereas synergy at the cell population level would depend on the heterogeneity in the susceptibilities across cells. To assess these possibilities, we applied our model next to available *in vitro* data such as the above and estimated the relative usage of the two pathways for entry in different cell lines (Methods). The different cell lines could be representative of the different cell and tissue types that could be infected *in vivo*. If the different cell population level. On the other hand, if all the cell lines used both pathways with similar relative pathway usage, synergy would arise mostly from the single cell level.

We collated *in vitro* data of SARS-CoV and SARS-CoV-2 pseudotyped virus infection of several different cell lines across studies [21,28] (S2 Table) and analysed it using our model. We found that the relative usage of the pathways was similar between SARS-CoV and SARS-CoV-2 but varied widely across cell types (Fig 6). For instance, with 293T cells expressing ACE2, entry was nearly exclusively via the Cathepsin B/L pathway. With Vero cells expressing TMPRSS2, entry of SARS-CoV-2 occurred ~65% of the time via the TMPRSS2 pathway and ~35% of the time through the Cathepsin B/L pathway. The corresponding figures were ~60% and ~40% for SARS-CoV. The latter were also similar to SARS-CoV entry into HeLa cells expressing ACE2 and TMPRSS2. With Caco cells, the relative usage of the TMPRSS2 pathway increased further to ~85% for both SARS-CoV and SARS-CoV-2 and culminated at nearly 100% usage of the TMPRSS2 pathway in Calu3 cells.



Fig 6. Relative usage of TMPRSS2 and Cathepsin B/L pathways during SARS-CoV and SARS-CoV-2 entry. The fraction of infection events unaffected either by TMPRSS2 or Cathepsin B/L inhibitors in excess were used to estimate the relative entry pathway usage in different cells lines. Data and sources are in S2 Table. The analysis follows the formalism outlined in the Methods.

This spectrum of the relative usage of the pathways across cell types is striking and spans the range from exclusive usage of one pathway to the nearly even usage of both. If this spectrum were indicative of the usage *in vivo*, which remains to be identified, it would imply that combination therapy targeting the two pathways would benefit from synergy arising from both the single cell level and the cell population level.

Discussion

The unprecedented scale of the COVID-19 pandemic has spurred urgent efforts to develop drugs and vaccines against SARS-CoV-2. Targeting host proteases required for SARS-CoV-2 entry into target cells appears a promising option and is under clinical investigation. Here, we identified unexpected synergy between drugs that target the serine protease TMPRSS2 and the cysteine proteases Cathepsin B/L, which offer alternative and independent pathways for SARS-CoV-2 entry. Using mathematical modelling of SARS-CoV-2 entry via the two pathways, we showed that the synergy is novel and arises from effects both at the single cell level and the cell population level. We found analysing several reported in vitro experimental observations that the synergy was evident not only with SARS-CoV-2 but also with the closely related and much more extensively studied SARS-CoV infection, which uses the same entry pathways, and was observed across drugs and cell lines, suggesting that the synergy is a robust phenomenon. Further, we estimated the relative usage of the two pathways and found it to vary widely across cell types, suggesting that synergy from the single cell and cell population levels is likely to arise *in vivo*, given the broad tissue tropism already identified for SARS-CoV-2 infection [21,44]. Drugs targeting TMRPSS2, namely, camostat mesylate and nafamostat mesylate, as well as drugs targeting the Cathepsin B/L pathway, such as hydroxychloroquine,

are in clinical trials for SARS-CoV-2 treatment [3,17]. Exploiting the synergy may maximize the impact of such drugs and improve the chances of curing SARS-CoV-2 infection.

Synergy can be of particular importance in the use of repurposed drugs. Because repurposed drugs are designed primarily for other, typically unrelated targets, they may display suboptimal efficacies against their newly intended targets. Indeed, a recent analysis of SARS-CoV-2 dynamics in 13 patients treated with different repurposed drugs found that the drug efficacies were in the range of 20–70% and were far below the minimum of 80% that would be required for effective treatment after the onset of symptoms [58]. Increasing drug levels to achieve desired efficacies is likely to be limited by toxicities. Synergy here could be particularly helpful for it would allow the realization of the desired efficacies at drug concentrations below what would be required had the drugs acted independently. Thus, the synergy between TMPRSS2 inhibitors and Cathepsin B/L inhibitors we unravelled may improve their deployability against SARS-CoV-2.

The synergy we elucidated is unconventional and arises because two independent entry pathways are accessible to SARS-CoV-2. Classically, synergy is thought to arise between drugs when one of the drugs potentiates the other through interactions between their targets or downstream pathways [40,41,43,50,52,59]. No such interaction is evident here. The synergy here arises because blocking both pathways is necessary for preventing entry and can only be achieved when both the drugs are used. Such synergy has been recognized earlier in the context of biochemical pathways where the same pathway can be triggered by two upstream stimuli independently [40]. Drugs that block the two stimuli, seemingly independent targets, display synergy, in a manner similar to what we have elucidated here. In addition, synergy arises in our case from the variations in the expression levels of the two proteases across cells. Because of the variations, some cells predominantly use one entry pathway and others the other pathway. Thus, protecting the entire cell population is achieved efficaciously when both the drugs are used. Such cell population level synergy has been argued to explain the observed synergy between drugs targeting hepatitis C virus entry and those targeting other steps of the hepatitis C virus lifecycle including type I interferons [55,60]. The overall synergy seen here is thus a convolution of the synergy at the single cell level as well as that at the cell population level. To our knowledge, such multi-level synergy between drugs has not been recognized earlier.

The multi-level synergy implies that the use of the drugs in combination may address redundancies at multiple levels that sustain the infection. The redundancy at the level of protease usage for infecting individual cells is well recognized [21,28]. The combination would block both TMPRSS2 and Cathepsin B/L, addressing the redundancy and thereby protecting cells. Redundancy may also arise at the cell population level. The cell tropism of SARS-CoV-2 in humans is only beginning to be identified and is speculated to be wider than SARS-CoV [46]. The expression level of TMPRSS2 varies across tissues in humans and mice [45–47]. It is possible thus that in some tissues, with high TMPRSS2 expression, SARS-CoV-2 predominantly uses TMPRSS2 for entry, whereas in others, with low TMPRSS2 expression, it uses Cathepsin B/L. Indeed, we estimated the relative usage of the entry pathways and found it to vary widely across cell lines, possibly indicative of the heterogeneity that may be expected *in vivo*. Protecting all the target tissues is thus only possible with the combination. If the extent of these redundancies *in vivo*, which remains to be ascertained, is high, the extent of synergy and hence the efficacy of the combination would be high.

Recently, a trade-off between synergy and efficacy has been identified in a variety of synergistic drug combinations, including those used in HIV, hepatitis C and cancer treatments [52,54]. The trade-off implies that as the efficacy of a combination increases beyond a point, the synergy decreases, thereby limiting the synergy that can be realized given a desired efficacy. This trade-off originates from the classical drug interactions underlying synergy, where, for instance, one drug potentiates the other in blocking different steps of the same pathway [52]. If one of the drugs is used at such high efficacies that the pathway is nearly completely blocked, then little room is left for the other drug to act and no synergy is possible. This trade-off is not expected to constrain the combination of TMPRSS2 and Cathepsin B/L inhibitors because the two target independent pathways and not different steps in the same pathway. Indeed, our model predictions showed that the synergy would increase with efficacy, implying a gain in synergy with increasing drug concentrations. The synergy may be compromised, however, and become constrained by the trade-off if one of the drugs were to exhibit activity against both the pathways, as has been suggested with the drug TLCK [28].

Mathematical models have been developed earlier to describe receptor-dependent virus entry, with a focus on defining the entry requirements of viruses and quantifying targets of entry inhibitors and vaccines [55,61–70]. The models, however, have focussed typically on a single entry pathway, although multiple proteins may have been involved in mediating entry through the pathway. For instance, models of HIV-1 entry, which requires the binding of the HIV-1 envelope gp120 to the cell surface receptor CD4 and the chemokine receptor CCR5 (or CXCR4), have focused on estimating the number of gp120 trimers that must engage with CD4 and CCR5 for successful entry [62-64,67]. Similarly, models of hepatitis C virus entry have focused on the number of viral envelope proteins E2 that must bind with the cell surface receptors CD81 for successful entry [61,65]. (In one study, the presence of another cell surface molecule SR-BI has been shown to aid hepatitis C virus entry, but entry without CD81 binding has not been demonstrated [65].) Accordingly, the models rarely admitted multiple entry pathways. Our model considered two independent entry pathways, allowing the elucidation of the synergy achieved by blocking them both. Furthermore, by using suitably designed limiting or extreme scenarios, our analysis brought out conceptually the origins of the synergy both at the single cell and the cell population levels. The resulting findings are likely to apply beyond SARS-CoV-2 because the entry of several other viruses, including other coronaviruses such as SARS-CoV, MERS-CoV, and influenza viruses [23], occurs similarly, via two independent pathways.

Furthermore, our model was able to estimate the relative usage of the two pathways across different cell lines by analysis of available *in vitro* data. We found that the relative usage of the pathways was similar between SARS-CoV-2 and SARS-CoV but varied widely across cell lines, with the usage of the TMPRSS2 pathway varying from ~0% to ~100% on average depending on the cell line. The variability could be due to the different expression levels of the proteases across the cell lines [45–47]. The variability implies that synergy both at the single cell and cell population may be expected *in vivo* from a drug combination targeting the two pathways.

Given a desired efficacy of the combination, arising from considerations, for instance, of achieving a desired decline in viral load [58], optimal drug dosages can be estimated that would maximize synergy while ensuring efficacy, as has been demonstrated in other settings [40,43,52,54,71]. Our model provides a framework that can be readily applied to predict these optimal dosages. The optimization would require knowledge of the distributions of the proteases across cells and tissue types targeted by SARS-CoV-2, which is beginning to be acquired [45,46]. We expect our study to prompt further investigations into the usage of the two pathways by SARS-CoV-2 *in vivo*.

Studies on other coronavirus infections of mice have remained inconclusive on the synergy between TMRPSS2 and Cathepsin B/L inhibitors. In some studies on SARS-CoV, synergy was not observed in mice although it was evident *in vitro* [72,73]. Studies with the coronavirus HCoV229E, among the viruses causing common cold, attributed such differences between *in vitro* and *in vivo* settings to mutations associated with the laboratory adaptation of the virus to

cell lines [74,75]. The mutations have been argued to allow Cathepsin B/L usage *in vitro*, while the *in vivo* strains predominantly use TMPRSS2. Yet, successful SARS-CoV infection of TMPRSS2 knockout mice was observed, demonstrating the ability of SARS-CoV and suggesting the ability of SARS-CoV-2 to use Cathepsin B/L *in vivo* [72]. Besides, the proposed laboratory adaptation appears unlikely to have happened with SARS-CoV-2 because the virus isolated from infected individuals was used in the experiments with minimal *in vitro* passaging [21]. Yet, synergy has been seen *in vitro* between camostat mesylate and E-64D with SARS-CoV-2 [21], suggesting that such synergy may also occur *in vivo*.

We note that in addition to TMPRSS2 and Cathepsin B/L, with SARS-CoV-2, unlike SARS-CoV, the use of furin for S protein activation has been suggested [20,46,76,77]. In a recent study, the role of furin in S protein activation of SARS-CoV-2 was argued to be similar to that in MERS-CoV, where furin cleaves S at the S1/S2 site, following which TMPRSS2 cleavage of S is required for entry through the fusion pathway [77]. Thus, furin does not appear to replace TMPRSS2, but instead places an additional requirement, akin to ACE2 binding, for entry through the TMPRSS2 pathway. Further, eliminating the furin cleavage site did not affect entry through the Cathepsin B/L, or endocytosis, pathway. Our prediction of the synergy between TMPRSS2 and Cathepsin B/L inhibitors is thus independent of the role of furin. Further, a recent study has argued that SARS-CoV-2 may utilize other TMPRSS2-related serine proteases to enter target cells with varying efficiencies [78]. Camostat mesylate was found to block all these proteases. Whether other inhibitors of TMPRSS2 also block related proteases remains to be ascertained. Combining Cathepsin B/L inhibitors with highly specific TMPRSS2 inhibitors may permit virus entry via TMPRSS2-related proteases, compromising therapy and limiting the synergy with Cathepsin B/L inhibitors.

In summary, our study identified unexpected synergy between TMPRSS2 and Cathepsin B/ L inhibitors against SARS-CoV-2 infection, highlighted its existence in available *in vitro* studies, and elucidated its novel origins at the single cell and cell population level using mathematical modelling. Our findings may have implications for the treatment of COVID-19 using agents blocking host proteases required for SARS-CoV-2 entry. Although we focussed on TMPRSS2 and Cathepsin B/L, our findings apply more broadly to combinations of agents that block the two independent pathways, fusion and endocytosis, of SARS-CoV-2 entry into target cells.

Methods

Model of SARS-CoV-2 entry via TMPRSS2 and Cathepsin B/L mediated pathways

We considered *in vitro* experiments where target cells expressing TMPRSS2 and Cathepsin B/ L were exposed to virions or pseudotyped viral particles bearing the spike protein, S. Experiments were performed in the absence or presence of inhibitors of TMPRSS2 and Cathepsin B/ L and the fraction of infection events measured post virus exposure. We developed the model below to describe the ensuing viral dynamics.

Viral dynamics. We recognized that the expression levels of the proteases would vary across cells and affect viral entry into the cells. Following previous approaches [55,61], we divided the cells into subpopulations based on the expression levels of the proteases. We defined T_{tc} as the subpopulation of target cells expressing TMPRSS2 in a narrow range Δn_t around n_t copies per cell and Cathepsin B/L in a narrow range Δn_c around n_c copies per cell. We described the relative susceptibility of these cells to virus entry through the TMPRSS2

pathway, S_t , and the Cathepsin B/L pathway, S_c , using Hill functions $S_t = \frac{(n_t)^{h_t}}{(n_t^{50})^{h_t} + (n_t)^{h_t}}$ and

 $S_c = \frac{(n_c)^{h_c}}{(n_c^{50})^{h_c}+(n_c)^{h_c}}$, where h_t and h_c were the Hill coefficients, and n_t^{50} and n_c^{50} were protease expression levels at which $S_t = 0.5$ and $S_c = 0.5$, respectively. Such Hill functional forms have been used to describe the effects of other proteases [79]. The overall susceptibility of the cells, due to entry through both the pathways, was $S_{tc} = S_t + S_c - S_t S_c$. In the presence of a TMPRSS2 inhibitor, a Cathepsin B/L inhibitor, or both, the susceptibilities were lowered to $S_{tc}(D_T)$, $S_{tc}(D_C)$, and $S_{tc}(D_T, D_C)$, respectively, where D_T and D_C were the concentrations of the TMPRSS2 and Cathepsin B/L inhibitors. The following equations described the ensuing viral dynamics:

$$\frac{dT_{tc}}{dt} = (\lambda - \mu)T_{tc} - kS_{tc}T_{tc}V; \ t = 1, 2, \dots, M; \ c = 1, 2, \dots, N$$
(1)

$$\frac{dI_{tc}}{dt} = kS_{tc}T_{tc}V - \delta I_{tc}; \ t = 1, 2, \dots, M; \ c = 1, 2, \dots, N$$
(2)

$$\frac{dV}{dt} = p \sum_{c=1}^{N} \sum_{t=1}^{M} I_{tc} - c_{V} V$$
(3)

Here, target cells proliferate and die with rate constants λ and μ , respectively. Incorporating the carrying capacity of the cell culture did not affect our results (S4 Fig). k is the second order rate constant of the infection by free virions, V, of target cells expressing excess TMPRSS2, Cathepsin B/L, or both, so that entry is not limited by the proteases, *i.e.*, for cells with the susceptibility $S_{tc} = 1$. k is reduced for cells T_{tc} by the factor S_{tc} , the relative susceptibility. We recognize that the susceptibility could be altered by variations in the expression levels of other factors such as ACE2 or type I interferon-stimulated proteins, as seen in measurements [45,46] and indicated by the substantial variability in the entry efficiency across different cell lines [21,49,80]. Our focus here is on the relative entry efficiency through the two protease-activated pathways. The susceptibilities are thus relative to the maximum entry efficiency when factors other than TMPRSS2 and Cathepsin B/L are limiting. Successful infection leads to the production of the corresponding infected cells I_{tc} . We neglected the proliferation of infected cells following previous models [61,81]. Infected cells die with a rate constant δ . Free virions are produced from infected cells at the rate p per cell and cleared with the rate constant c_V . We focussed on experiments with pseudotyped viruses, with I_{tc} representing cells into which successful virus entry has occurred. These pseudotyped viruses are replication incompetent and do not produce new virions. We therefore neglected progeny virion production (p = 0).

Distribution of protease expression levels across cells. We assumed that at the start of infection, the fraction, $\phi_{tc} = \psi_{tc}(n_t, n_c) \Delta n_t \Delta n_c$, of cells belonging to the subpopulation T_{tc} expressing the proteases in the narrow range Δn_t and Δn_c around n_t and n_c , respectively, was determined by the joint distribution, $\psi_{tc}(n_t, n_c) = \psi_t(n_t)\psi_c(n_c)$, where $\psi_t(n_t) = \frac{1}{n_t \sigma_t \sqrt{2\pi}} e^{-\frac{(\ln n_t - \overline{n_t})^2}{2\sigma_c^2}}$ were the independent log-normal distributions of the expression

levels of TMPRSS2 and Cathepsin B/L, respectively, across cells with $\overline{n_t}$ and $\overline{n_c}$ the associated means, and σ_t and σ_c the standard deviations. We divided the target cells into $M \times N$ subpopulations, so that t = 1, 2, ..., M and c = 1, 2, ..., N.

Effect of drugs. We assumed that drugs bound their target proteases and prevented their functioning. The susceptibility of cells would thus be reduced and be determined by the protease molecules unbound to the inhibitors. In the presence of a TMPRSS2 inhibitor, we let the abundance of free TMPRSS2, denoted n_t^f , follow $n_t^f = \frac{\gamma_T}{\gamma_T + D_T} n_t$. The susceptibility of cells T_{tc} to

infection through the TMPRSS2 pathway reduced to $S_t(D_T) = \frac{(n_t^f)^{h_t}}{(n_t^{50})^{h_t} + (n_t^f)^{h_t}}$, and the overall susceptibility became $S_{tc}(D_T) = S_t(D_T) + S_c - S_t(D_T)S_c$. Similarly, in the presence of a Cathepsin B/L inhibitor, we let the abundance of free Cathepsin B/L, n_c^f , available for mediating virus entry, follow $n_c^f = \frac{\gamma_C}{\gamma_C + D_C} n_c$. The susceptibilities accordingly reduced to $S_c(D_C) = \frac{(n_c^f)^{h_c}}{(n_c^{50})^{h_c} + (n_c^f)^{h_c}}$, and $S_{tc}(D_C) = S_t + S_c(D_C) - S_t S_c(D_C)$. With both the drugs used simultaneously, the susceptibilities were $S_{tc}(D_T, D_C) = S_t(D_T) + S_c(D_C) - S_t(D_T)S_c(D_C)$. Here, D_T was the concentration of the TMPRSS2 inhibitor, D_C that of the Cathepsin B/L inhibitor, and γ_T and γ_C were the inhibition constants representing drug levels that reduced the respective free protease levels by half. These expressions for the free protease numbers followed dose-response relationships that could be derived mechanistically assuming reaction equilibrium of drug-protease binding and species balance constraints [61].

Synergy. We solved Eqns. [1–3] to predict the number of infected cells at day 1 post-infection in each subpopulation in the absence of drugs, denoted I_{tc} , in the presence of a TMPRSS2 inhibitor, $I_{tc}(D_T)$, in the presence of a Cathepsin B/L inhibitor, $I_{tc}(D_C)$, and in the presence of both inhibitors, $I_{tc}(D_T,D_C)$. From these populations, we estimated the total fractions unaffected

by the drugs individually and together as $f^u(D_T) = \frac{\sum_{c=1}^N \sum_{t=1}^M I_{tc}(D_T)}{\sum_{c=1}^N \sum_{t=1}^M I_{tc}(D_C)} = \frac{\sum_{c=1}^N \sum_{t=1}^M I_{tc}(D_C)}{\sum_{c=1}^N \sum_{t=1}^M I_{tc}(D_C)}$

The fully and together as
$$f^{\mu}(D_T) = \frac{1}{\sum_{c=1}^{N} \sum_{t=1}^{M} I_{tc}}, f^{\mu}(D_C) = \frac{1}{\sum_{c=1}^{N} \sum_{t=1}^{M} I_{tc}},$$

and $f^{u}(D_{T}, D_{C}) = \frac{\sum_{t=1}^{N} \sum_{t=1}^{M} I_{tc}(D_{T}, D_{C})}{\sum_{c=1}^{N} \sum_{t=1}^{M} I_{tc}}$. The expected fraction unaffected assuming Bliss inde-

pendence was thus $f^{u}_{Bliss}(D_T, D_C) = f^{u}(D_T)f^{u}(D_C)$, and the extent of Bliss synergy, $\beta_{Bliss} = f^{u}_{Bliss}(D_T, D_C) - f^{u}(D_T, D_C)$. Thus,

$$\beta_{Bliss} = \frac{\sum_{c=1}^{N} \sum_{t=1}^{M} I_{tc}(D_T)}{\sum_{c=1}^{N} \sum_{t=1}^{M} I_{tc}} \times \frac{\sum_{c=1}^{N} \sum_{t=1}^{M} I_{tc}(D_C)}{\sum_{c=1}^{N} \sum_{t=1}^{M} I_{tc}} - \frac{\sum_{c=1}^{N} \sum_{t=1}^{M} I_{tc}(D_T, D_C)}{\sum_{c=1}^{N} \sum_{t=1}^{M} I_{tc}}.$$
(4)

To estimate the extent of synergy arising from the effects of the drugs at the single cell level, β_{Bliss}^{cell} , we performed the same calculations as above but assuming a homogeneous cell population.

Identifying the synergy at the cell population level was more involved. It required deriving ways to decouple single cell level synergy from the cell population level synergy. We accomplished it for single round infection assays, which we describe in <u>S3 Text</u>. Accordingly, the cell population synergy was given by the following expression:

$$\beta_{Bliss}^{popn} = \frac{1}{\left[\sum_{c=1}^{N}\sum_{t=1}^{M}\phi_{tc}(S_{t}+S_{c}-S_{t}S_{c})\right]^{2}} \begin{bmatrix} \left[\sum_{c=1}^{N}\sum_{t=1}^{M}\phi_{tc}(S_{t}-S_{t}(D_{T}))(1-S_{c})\right] \left[\sum_{c=1}^{N}\sum_{t=1}^{M}\phi_{tc}(S_{c}-S_{c}(D_{C}))(1-S_{t})\right] \\ -\sum_{c=1}^{N}\sum_{t=1}^{M}\phi_{tc}^{2}(S_{t}-S_{t}(D_{T}))(1-S_{t})(S_{c}-S_{c}(D_{C}))(1-S_{c}) \end{bmatrix} \end{bmatrix}.$$
(5)

Estimates of the relative usage of the entry pathways

We recognized from our analysis above that data from single round infection assays or, equivalently, assays using pseudotyped viruses could be used to estimate the relative usage of the two pathways for entry. We achieved this by linking the fraction of infection events unaffected by drugs and the mean susceptibilities of the populations as follows. Recall that the susceptibility of the cell subpopulation T_{tc} is $S_{tc} = S_t + S_c - S_t S_c$, the latter equal to the fraction of the subpopulation infected on average in the absence of drugs in a single round assay. Accordingly, the total fraction of cells infected in the absence of drugs would be

$$f^{\text{inf}} = \sum_{c=1}^{N} \sum_{t=1}^{M} \phi_{tc} S_{tc} = \sum_{c=1}^{N} \sum_{t=1}^{M} \phi_{tc} (S_t + S_c - S_t S_c) = \overline{S_t} + \overline{S_c} - \overline{S_t} \cdot \overline{S_c}, \text{ where we defined the}$$

mean susceptibilities of the populations to entry through the TMPRSS2 pathway,

$$\overline{S_t} = \sum_{t=1}^{M} \phi_t S_t$$
, and that through the Cathepsin B/L pathway, $\overline{S_c} = \sum_{c=1}^{N} \phi_c S_c$, and used the inde-

pendence of the distributions of the expression levels of the proteases, $\phi_{tc} = \phi_t \phi_c$.

Following the arguments above, the fractions infected in the presence of drugs were similarly $f^{\text{inf}}(D_T) = \overline{S_t(D_T)} + \overline{S_c} - \overline{S_t(D_T)} \cdot \overline{S_c}$, $f^{\text{inf}}(D_C) = \overline{S_t} + \overline{S_c(D_C)} - \overline{S_t} \cdot \overline{S_c(D_C)}$, and $f^{\text{inf}}(D_T, D_C) = \overline{S_t(D_T)} + \overline{S_c(D_C)} - \overline{S_t(D_T)} \cdot \overline{S_c(D_C)}$, where $\overline{S_t(D_T)}$ and $\overline{S_c(D_C)}$ were the mean susceptilities of the populations in the presence of a TMPRSS2 inhibitor and a Cathepsin B/L inhibitor, respectively. These expressions yielded the fractions unaffected by drugs as:

$$f^{u}(D_{T}) = \frac{\overline{S_{t}(D_{T})} + \overline{S_{c}} - \overline{S_{t}(D_{T})} \cdot \overline{S_{c}}}{\overline{S_{t}} + \overline{S_{c}} - \overline{S_{t}} \cdot \overline{S_{c}}}$$
(6)

$$f^{u}(D_{c}) = \frac{\overline{S_{t}} + \overline{S_{c}(D_{c})} - \overline{S_{t}} \cdot \overline{S_{c}(D_{c})}}{\overline{S_{t}} + \overline{S_{c}} - \overline{S_{t}} \cdot \overline{S_{c}}}$$
(7)

$$f^{\mu}(D_T, D_C) = \frac{\overline{S_t(D_T)} + \overline{S_c(D_C)} - \overline{S_t(D_T)} \cdot \overline{S_c(D_C)}}{\overline{S_t} + \overline{S_c} - \overline{S_t} \cdot \overline{S_c}}$$
(8)

The above (Eqs [6–8]) were three equations in the four unknowns $\overline{S_t}$, $\overline{S_c}$, $\overline{S_t(D_T)}$, and $\overline{S_c(D_C)}$, and needed an additional constraint for solution. Our goal was to estimate the ratio of $\overline{S_t}$ and $\overline{S_c}$ using the reported measurements of $f^{\mu}(D_T)$, $f^{\mu}(D_C)$, and $f^{\mu}(D_T, D_C)$. For this, experiments with saturating drug levels offered a strategy. We considered experiments where the drug levels used were high enough that they blocked their respective pathways nearly fully. In other words, $\overline{S_t(D_T)} \sim \overline{S_c(D_C)} \sim 0$. A test of this criterion was that a nearly complete block of infection occurred when the drugs were used in combination because the criterion would imply $f^{\mu}(D_T, D_C) \sim 0$ (Eq [8]). Under these circumstances, Eqs [6] and [7] became $f^u(D_T) \approx \frac{\overline{S_c}}{\overline{S_t + \overline{S_c} - \overline{S_t} \cdot \overline{S_c}}$, dividing which yielded $f^u(D_C)/f^u(D_T) = \overline{S_t}/\overline{S_c}$, so that the relative usage of the TMPRSS2 pathway was given by

$$\frac{\overline{S_t}}{\overline{S_t} + \overline{S_c}} = \frac{f^u(D_C)}{f^u(D_C) + f^u(D_T)}.$$
(9)

Data

We considered experiments were different cell lines are exposed to SAR-CoV or SARS-CoV-2 pseudotyped virions in the presence or absence of TMPRSS2 and Cathepsin B/L inhibitors, and the fraction of infection events unaffected by the drugs was estimated [21,28]. We extracted the data reported using Engauge Digitizer 12.1 and estimated the extent of synergy using the Bliss index. The data is presented in Table 1 and S2 Table. The cell lines and drugs used are mentioned in the Results section and Tables.

Model calculations and parameters

The model equations were solved using programs written in MATLAB (MathWorks, Natick, MA) and Mathematica. We chose the target cell proliferation and death rate constants representative of Vero cells, typically used for studying SARS-CoV-2 infections in vitro. These have been estimated to be 0.77 d⁻¹ and 0.22 d⁻¹ [82,83]. The infected cell death rate and the virion clearance rate constants have been chosen to apply to SARS-CoV-2 infections. These have been set to 0.53 d⁻¹ and 10 d⁻¹ in earlier studies [58]. The virion production rate per infected cell was set to zero mimicking pseudo-typed virus infection. The initial target cell and virion populations, the infection rate constant and Hill coefficients in the susceptibility expressions were drawn from studies on other viral infection kinetics, likely typical of *in vitro* assays [55,84]. The means and standard deviations of protease levels as well as the levels at which the susceptibility of cells to entry is 0.5 remain unknown. These parameters were assumed, mimicking other viruses where available [55], and were varied over a large range to assess the sensitivity of our model predictions to uncertainties in their estimates. Our key conclusions were unaffected by these variations (S1 and S2 Figs). The infection and the extent of inhibition were assessed at 1 d post-infection, mimicking experimental protocols [21]. M and N have been set to be 40. Parameter values are listed in <u>S1 Table</u>. Any deviations from these values are mentioned in the figure captions.

Parameter sensitivity

We first tested the sensitivity of model predictions to parameter values by increasing or decreasing model parameter values by 2-fold from the baseline value one at time (S1 Fig). We found that the model predicted synergy over a broad range of parameter values (S1 Fig). We next performed global sensitivity analysis and computed partial rank correlation coefficients (PRCCs) of each model parameter [85]. The predictions of synergy were most sensitive to the mean and variance of the expression levels of the proteases and the expression levels defining half-maximal susceptibility (S2A Fig). Synergy diminished in parameter regimes where one entry pathway dominated the other (S2B Fig).

Supporting information

S1 Text. Analytical expression of synergy at the single cell level. (DOCX)

S2 Text. Synergy at the cell population level in the two-subpopulation scenario. (DOCX)

S3 Text. Model of single round infection assay. (DOCX)

S1 Table. Model parameters and their values. The meanings of the symbols are in the Methods. The effects of variations in these parameter values are in <u>S1 Fig</u>. (DOCX)

S2 Table. Relative usage of TMPRSS2 and Cathepsin B/L pathways of entry. We computed the relative pathway usage from the effects of the individual drugs targeting either TMPRSS2 or Cathepsin B/L pathways on overall infection. Infection was by viruses pseudotyped with either SARS-CoV-2 S or SARS-CoV S protein. The fraction uninhibited when both the drugs are used was nearly zero in all the cases considered, indicating that the criterion required for the application of our model was met (see Methods). The relative usage we estimated is presented in Fig 6.



S1 Fig. Sensitivity to model parameters. The Bliss synergy predicted by the full model by changing model parameter values from the baseline one at a time. The black bars are predictions using the baseline parameter values in <u>S1 Table</u>. The red and blue bars are predictions with parameter values 2-fold lower and higher than the baseline values, respectively. Other parameters values used in various bars: $\overline{n_t} = 9.5$ (red), 11.5 (black) and 13.5 (blue); $\overline{n_c} = 9.5$ (red), 11.5 (black) and 13.5 (blue). The drug concentrations used are $D_T/\gamma_T = D_C/\gamma_C = 10^3$. (TIF)

S2 Fig. Global sensitivity analysis. (A) Partial rank correlation coefficients (PRCCs) indicating the sensitivity of the model predictions of Bliss synergy to variations in model parameter values. The predicted Bliss synergy is most sensitive to the mean and variance of the expression levels of the proteases and the expression levels defining half-maximal susceptibility. The drug concentrations used are $D_T/\gamma_T = D_C/\gamma_C = 10^3$. PRCCs were computed from 2000 runs. $\overline{n_t}$ and $\overline{n_c}$ were varied between $\ln(\frac{10^5}{4})$ and $\ln(4 \times 10^5)$, and the other parameters were varied up to 4-fold above and below their respective baseline values. (B) The distribution of Bliss synergy predicted from 2000 runs in A. For this analysis, we adapted the MATLAB codes available on Prof. Denise Kirschner's website (http://malthus.micro.med.umich.edu/lab/usadata/). (TIF)

S3 Fig. Comparison between numerical predictions and analytical approximations of synergy. (A, B) The predicted Bliss synergy at the single cell level for varying TMPRSS2 expression and fixed ($n_c = 150,000$ copies/cell) Cathepsin B/L expression (A) and for varying Cathepsin B/L expression and fixed ($n_t = 100,000$ copies/cell) TMPRSS2 expression (C) at two different drug concentrations. Solid lines are predictions of the full model with homogeneous protease expression across cells (same as Fig 4). Dashed lines are predictions of the analytical expression $\beta_{Bliss}^{cell} = (S_t - S_t(D_T))(S_c - S_c(D_C))/(S_t + S_c - S_tS_c)^2$ derived in S1 Text. Other parameters are the same as in Fig 4. (TIF)

S4 Fig. Carrying capacity of target cells did not affect predictions of overall Bliss synergy. (A, B) The predicted Bliss synergy as function of concentrations of TMPRSS2 inhibitor, D_T/γ_T , and Cathepsin B/L, D_C/γ_C , with (black line) and without logistic growth (coloured lines).

The equation
$$\frac{dT_{tc}}{dt} = (\lambda - \mu)T_{tc} \left(1 - \frac{\sum_{c=1}^{N} \sum_{t=1}^{M} T_{tc}}{T_{max}}\right) - kS_{tc}T_{tc}V; \ t = 1, 2, \dots, M; \ c = 1, 2, \dots, M;$$

 $1, 2, \ldots, N$ was used instead of Eq (1) in the main text to mimic logistic growth. Here, T_{max} is

the carrying capacity of the cell culture, T(0) is the initial number of target cells, $D_T/\gamma_T = D_C/\gamma_C = D/\gamma$, and other parameters are the same as in Fig.3. Note that $T_{\text{max}} \rightarrow \infty$ reduces the equation above to Eq (1).

(TIF)

Author Contributions

Conceptualization: Pranesh Padmanabhan, Narendra M. Dixit.

Formal analysis: Pranesh Padmanabhan, Rajat Desikan, Narendra M. Dixit.

Funding acquisition: Narendra M. Dixit.

Visualization: Pranesh Padmanabhan, Rajat Desikan.

Writing - original draft: Pranesh Padmanabhan, Narendra M. Dixit.

Writing - review & editing: Pranesh Padmanabhan, Rajat Desikan, Narendra M. Dixit.

References

- 1. WHO. Situation Report—206. 13 Aug 2020. Available: https://www.who.int/docs/default-source/ coronaviruse/situation-reports/20200520-covid-19-sitrep-121.pdf?sfvrsn=c4be2ec6_2
- de Wit E, van Doremalen N, Falzarano D, Munster VJ. SARS and MERS: Recent insights into emerging coronaviruses. Nat Rev Microbiol. 2016; 14: 523–534. https://doi.org/10.1038/nrmicro.2016.81 PMID: 27344959
- 3. Sanders JM, Monogue ML, Jodlowski TZ, Cutrell JB. Pharmacologic treatments for coronavirus disease 2019 (COVID-19): A review. JAMA. https://doi.org/10.1001/jama.2020.6019 PMID: 32282022
- Amanat F, Krammer F. SARS-CoV-2 vaccines: Status report. Immunity. 2020; 52: 583–589. https://doi. org/10.1016/j.immuni.2020.03.007 PMID: 32259480
- Li G, De Clercq E. Therapeutic options for the 2019 novel coronavirus (2019-nCoV). Nat Rev Drug Discov. 2020; 19: 149–150. https://doi.org/10.1038/d41573-020-00016-0 PMID: 32127666
- Liu C, Zhou Q, Li Y, Garner LV, Watkins SP, Carter LJ, et al. Research and development on therapeutic agents and vaccines for covid-19 and related human coronavirus diseases. ACS Cent Sci. 2020; 6: 315–331. https://doi.org/10.1021/acscentsci.0c00272 PMID: 32226821
- Zhou Y, Hou Y, Shen J, Huang Y, Martin W, Cheng F. Network-based drug repurposing for novel coronavirus 2019-nCoV/SARS-CoV-2. Cell Discov. 2020; 6: 14. <u>https://doi.org/10.1038/s41421-020-0153-3</u> PMID: 32194980
- "Solidarity" clinical trial for COVID-19 treatments. Available: https://www.who.int/emergencies/ diseases/novel-coronavirus-2019/global-research-on-novel-coronavirus-2019-ncov/solidarity-clinicaltrial-for-covid-19-treatments
- 9. Dittmar M, Lee JS, Whig K, Segrist E, Li M, Jurado K, et al. Drug repurposing screens reveal FDA approved drugs active against SARS-Cov-2. bioRxiv. https://doi.org/10.1101/2020.06.19.161042
- Beigel JH, Tomashek KM, Dodd LE, Mehta AK, Zingman BS, Kalil AC, et al. Remdesivir for the treatment of covid-19—preliminary report. N Engl J Med. 2020. https://doi.org/10.1056/NEJMoa2007764 PMID: 32445440
- Horby P, Lim WS, Emberson J, Mafham M, Bell J, Linsell L, et al. Effect of dexamethasone in hospitalized patients with COVID-19: Preliminary report. medRxiv. https://doi.org/10.1101/2020.06.22. 20137273 PMID: 32690491
- Ivashchenko AA, Dmitriev KA, Vostokova NV, Azarova VN, Blinow AA, Egorova AN, et al. Avifavir for treatment of patients with moderate covid-19: Interim results of a phase II/III multicenter randomized clinical trial. medRxiv. https://doi.org/10.1093/cid/ciaa1176 PMID: 32770240
- Liu J, Cao R, Xu M, Wang X, Zhang H, Hu H, et al. Hydroxychloroquine, a less toxic derivative of chloroquine, is effective in inhibiting SARS-CoV-2 infection in vitro. Cell Discov. 2020; 6: 16. <u>https://doi.org/10. 1038/s41421-020-0156-0 PMID: 32194981</u>
- National Task Force for COVID-19, ICMR, Government of India. Advisory on the use of Hydroxychloroquine as prophylaxis for SARS-CoV-2 infection. 21 Mar 2020. Available: https://www.mohfw.gov.in/pdf/ AdvisoryontheuseofHydroxychloroquinasprophylaxisforSARSCoV2infection.pdf

- Hoffmann M, Mösbauer K, Hofmann-Winkler H, Kaul A, Kleine-Weber H, Krüger N, et al. Chloroquine does not inhibit infection of human lung cells with SARS-CoV-2. Nature. 2020; 1–5. https://doi.org/10. 1021/acschembio.0c00553 PMID: 32786258
- Ou T, Mou H, Zhang L, Ojha A, Choe H, Farzan M. Hydroxychloroquine-mediated inhibition of SARS-CoV-2 entry is attenuated by TMPRSS2. bioRxiv. https://doi.org/10.1101/2020.07.22.216150
- Zhu R, Gao Y, Robert S-H, Gao J, Yang S, Zhu C. Systematic review of the registered clinical trials for coronavirus disease 2019 (COVID-19). J Transl Med. 18: 274. <u>https://doi.org/10.1186/s12967-020-02442-5 PMID: 32631442</u>
- Le TT, Andreadakis Z, Kumar A, Roman RG, Tollefsen S, Saville M, et al. The COVID-19 vaccine development landscape. Nat Rev Drug Discov. 2020; 19: 305–6. https://doi.org/10.1038/d41573-020-00073-5 PMID: 32273591
- Gordon DE, Jang GM, Bouhaddou M, Xu J, Obernier K, White KM, et al. A SARS-CoV-2 protein interaction map reveals targets for drug repurposing. Nature. 2020; 1–13. <u>https://doi.org/10.1038/s41586-020-</u> 2286-9 PMID: 32353859
- Walls AC, Park Y-J, Tortorici MA, Wall A, McGuire AT, Veesler D. Structure, function, and antigenicity of the SARS-CoV-2 spike glycoprotein. Cell. 2020; 180: 1–12. <u>https://doi.org/10.1016/j.cell.2019.12.</u> 001 PMID: 31836194
- Hoffmann M, Kleine-Weber H, Schroeder S, Krüger N, Herrler T, Erichsen S, et al. SARS-CoV-2 cell entry depends on ACE2 and TMPRSS2 and is blocked by a clinically proven protease inhibitor. Cell. 2020; 181: 1–10. https://doi.org/10.1016/j.cell.2020.03.025 PMID: 32243785
- Hoffmann M, Hofmann-Winkler H, Pöhlmann S. Priming time: How cellular proteases arm coronavirus spike proteins. In: Böttcher-Friebertshäuser E, Garten W, Klenk HD, editors. Activation of Viruses by Host Proteases. Springer International Publishing; 2018. pp. 71–98.
- Karakus U, Pohl MO, Stertz S. Breaking the convention: sialoglycan variants, coreceptors, and alternative receptors for influenza A virus entry. J Virol. 2020; 94. https://doi.org/10.1128/JVI.01357-19 PMID: 31776280
- Letko M, Marzi A, Munster V. Functional assessment of cell entry and receptor usage for SARS-CoV-2 and other lineage B betacoronaviruses. Nat Microbiol. 2020; 5: 562–569. <u>https://doi.org/10.1038/</u> s41564-020-0688-y PMID: 32094589
- Ou X, Liu Y, Lei X, Li P, Mi D, Ren L, et al. Characterization of spike glycoprotein of SARS-CoV-2 on virus entry and its immune cross-reactivity with SARS-CoV. Nat Commun. 2020; 11: 1620. <u>https://doi.org/10.1038/s41467-020-15562-9 PMID: 32221306</u>
- Simmons G, Gosalia DN, Rennekamp AJ, Reeves JD, Diamond SL, Bates P. Inhibitors of cathepsin L prevent severe acute respiratory syndrome coronavirus entry. Proc Natl Acad Sci U S A. 2005; 102: 11876–11881. https://doi.org/10.1073/pnas.0505577102 PMID: 16081529
- Matsuyama S, Nagata N, Shirato K, Kawase M, Takeda M, Taguchi F. Efficient activation of the severe acute respiratory syndrome coronavirus spike protein by the transmembrane protease TMPRSS2. J Virol. 2010; 84: 12658–12664. https://doi.org/10.1128/JVI.01542-10 PMID: 20926566
- Kawase M, Shirato K, van der Hoek L, Taguchi F, Matsuyama S. Simultaneous treatment of human bronchial epithelial cells with serine and cysteine protease inhibitors prevents severe acute respiratory syndrome coronavirus entry. J Virol. 2012; 86: 6537–6545. <u>https://doi.org/10.1128/JVI.00094-12</u> PMID: 22496216
- 29. Glowacka I, Bertram S, Muller MA, Allen P, Soilleux E, Pfefferle S, et al. Evidence that TMPRSS2 activates the severe acute respiratory syndrome coronavirus spike protein for membrane fusion and reduces viral control by the humoral immune response. J Virol. 2011; 85: 4122–4134. <u>https://doi.org/10.1128/JVI.02232-10 PMID: 21325420</u>
- Shulla A, Heald-Sargent T, Subramanya G, Zhao J, Perlman S, Gallagher T. A transmembrane serine protease is linked to the severe acute respiratory syndrome coronavirus receptor and activates virus entry. J Virol. 2011; 85: 873–882. https://doi.org/10.1128/JVI.02062-10 PMID: 21068237
- Wu F, Zhao S, Yu B, Chen Y-M, Wang W, Song Z-G, et al. A new coronavirus associated with human respiratory disease in China. Nature. 2020; 579: 265–269. https://doi.org/10.1038/s41586-020-2008-3 PMID: 32015508
- Raj VS, Mou H, Smits SL, Dekkers DHW, Müller MA, Dijkman R, et al. Dipeptidyl peptidase 4 is a functional receptor for the emerging human coronavirus-EMC. Nature. 2013; 495: 251–254. <u>https://doi.org/ 10.1038/nature12005</u> PMID: 23486063
- Shirato K, Kawase M, Matsuyama S. Middle east respiratory syndrome coronavirus infection mediated by the transmembrane serine protease TMPRSS2. J Virol. 2013; 87: 12552–12561. https://doi.org/10. 1128/JVI.01890-13 PMID: 24027332

- Turner AJ, Hiscox JA, Hooper NM. ACE2: from vasopeptidase to SARS virus receptor. Trends Pharmacol Sci. 2004; 25: 291–294. https://doi.org/10.1016/j.tips.2004.04.001 PMID: 15165741
- 35. Gibo J, Ito T, Kawabe K, Hisano T, Inoue M, Fujimori N, et al. Camostat mesilate attenuates pancreatic fibrosis via inhibition of monocytes and pancreatic stellate cells activity. Lab Invest. 2005; 85: 75–89. https://doi.org/10.1038/labinvest.3700203 PMID: 15531908
- Hoffmann M, Schroeder S, Kleine-Weber H, Müller MA, Drosten C, Pöhlmann S. Nafamostat mesylate blocks activation of SARS-CoV-2: New treatment option for COVID-19. Antimicrob Agents Chemother. https://doi.org/10.1128/AAC.00754-20 PMID: 32312781
- **37.** Li Y-Y, Fang J, Ao G-Z. Cathepsin B and L inhibitors: a patent review (2010—present). Expert Opin Ther Pat. 2017; 27: 643–656. https://doi.org/10.1080/13543776.2017.1272572 PMID: 27998201
- Kramer L, Turk D, Turk B. The future of cysteine cathepsins in disease management. Trends Pharmacol Sci. 2017; 38: 873–898. https://doi.org/10.1016/j.tips.2017.06.003 PMID: 28668224
- Hu TY, Frieman M, Wolfram J. Insights from nanomedicine into chloroquine efficacy against COVID-19. Nat Nanotechnol. https://doi.org/10.1038/s41565-020-0674-9 PMID: 32203437
- 40. Fitzgerald JB, Schoeberl B, Nielsen UB, Sorger PK. Systems biology and combination therapy in the quest for clinical efficacy. Nat Chem Biol. 2006; 2: 458–466. https://doi.org/10.1038/nchembio817 PMID: 16921358
- Meyer CT, Wooten DJ, Lopez CF, Quaranta V. Charting the fragmented landscape of drug synergy. Trends Pharmacol Sci. 2020; 41: 266–280. https://doi.org/10.1016/j.tips.2020.01.011 PMID: 32113653
- Zimmermann GR, Lehár J, Keith CT. Multi-target therapeutics: when the whole is greater than the sum of the parts. Drug Discov Today. 2007; 12: 34–42. <u>https://doi.org/10.1016/j.drudis.2006.11.008</u> PMID: 17198971
- Meyer CT, Wooten DJ, Paudel BB, Bauer J, Hardeman KN, Westover D, et al. Quantifying drug combination synergy along potency and efficacy axes. Cell Syst. 2019; 8: 97–108.e16. <u>https://doi.org/10.1016/j.cels.2019.01.003</u> PMID: 30797775
- Lamers MM, Beumer J, Vaart J van der, Knoops K, Puschhof J, Breugem TI, et al. SARS-CoV-2 productively infects human gut enterocytes. Science. https://doi.org/10.1126/science.abc1669 PMID: 32358202
- 45. Sungnak W, Huang N, Bécavin C, Berg M, Queen R, Litvinukova M, et al. SARS-CoV-2 entry factors are highly expressed in nasal epithelial cells together with innate immune genes. Nat Med. 2020; 1–7. https://doi.org/10.1038/s41591-019-0740-8 PMID: 31932805
- Lukassen S, Chua RL, Trefzer T, Kahn NC, Schneider MA, Muley T, et al. SARS-CoV-2 receptor ACE2 and TMPRSS2 are primarily expressed in bronchial transient secretory cells. The EMBO Journal. 2020; 39: e105114. https://doi.org/10.15252/embj.20105114 PMID: 32246845
- Vaarala MH, Porvari KS, Kellokumpu S, Kyllönen AP, Vihko PT. Expression of transmembrane serine protease TMPRSS2 in mouse and human tissues. J Pathol. 2001; 193: 134–140. https://doi.org/10. 1002/1096-9896(2000)9999:9999<::AID-PATH743>3.0.CO;2-T PMID: 11169526
- Paoloni-Giacobino A, Chen H, Peitsch MC, Rossier C, Antonarakis SE. Cloning of the TMPRSS2 gene, which encodes a novel serine protease with transmembrane, LDLRA, and SRCR Domains and maps to 21q22.3. Genomics. 1997; 44: 309–320. https://doi.org/10.1006/geno.1997.4845 PMID: 9325052
- 49. Matsuyama S, Nao N, Shirato K, Kawase M, Saito S, Takayama I, et al. Enhanced isolation of SARS-CoV-2 by TMPRSS2-expressing cells. Proc Natl Acad Sci U S A. 2020; 117: 7001–7003. <u>https://doi.org/10.1073/pnas.2002589117</u> PMID: 32165541
- 50. Foucquier J, Guedj M. Analysis of drug combinations: current methodological landscape. Pharmacol Res Perspect. 2015; 3: e00149. https://doi.org/10.1002/prp2.149 PMID: 26171228
- Jilek BL, Zarr M, Sampah ME, Rabi SA, Bullen CK, Lai J, et al. A quantitative basis for antiretroviral therapy for HIV-1 infection. Nat Med. 2012; 18: 446–451. https://doi.org/10.1038/nm.2649 PMID: 22344296
- 52. Sen P, Saha A, Dixit NM. You cannot have your synergy and efficacy too. Trends Pharmacol Sci. 2019; 40: 811–817. https://doi.org/10.1016/j.tips.2019.08.008 PMID: 31610891
- Padmanabhan P, Garaigorta U, Dixit NM. Emergent properties of the interferon-signalling network may underlie the success of hepatitis C treatment. Nat Commun. 2014; 5: 1–9. <u>https://doi.org/10.1038/</u> ncomms4872 PMID: 24834957
- 54. Gupta V, Dixit NM. Trade-off between synergy and efficacy in combinations of HIV-1 latency-reversing agents. PLoS Comput Biol. 2018; 14: e1006004. <u>https://doi.org/10.1371/journal.pcbi.1006004</u> PMID: 29451894
- 55. Padmanabhan P, Dixit N. Modeling suggests a mechanism of synergy between hepatitis C virus entry inhibitors and drugs of other classes. CPT: Pharmacometrics Syst Pharmacol. 2015; 4: 445–453. https://doi.org/10.1002/psp4.12005 PMID: 26380153

- 56. Venugopal V, Padmanabhan P, Raja R, Dixit NM. Modelling how responsiveness to interferon improves interferon-free treatment of hepatitis C virus infection. PLoS Comput Biol. 2018; 14: e1006335. <u>https:// doi.org/10.1371/journal.pcbi.1006335</u> PMID: 30001324
- Raja R, Baral S, Dixit NM. Interferon at the cellular, individual, and population level in hepatitis C virus infection: Its role in the interferon-free treatment era. Immunol Rev. 2018; 285: 55–71. <u>https://doi.org/</u> 10.1111/imr.12689 PMID: 30129199
- Gonçalves A, Bertrand J, Ke R, Comets E, Lamballerie X de, Malvy D, et al. Timing of antiviral treatment initiation is critical to reduce SARS-CoV-2 viral load. CPT: Pharmacometrics Syst Pharmacol. <u>https:// doi.org/10.1002/psp4.12543</u> PMID: 32558354
- Baeder DY, Yu G, Hozé N, Rolff J, Regoes RR. Antimicrobial combinations: Bliss independence and Loewe additivity derived from mechanistic multi-hit models. Phil Trans R Soc B. 2016; 371: 20150294. https://doi.org/10.1098/rstb.2015.0294 PMID: 27160596
- Xiao F, Fofana I, Thumann C, Mailly L, Alles R, Robinet E, et al. Synergy of entry inhibitors with directacting antivirals uncovers novel combinations for prevention and treatment of hepatitis C. Gut. 2015; 64: 483–494. https://doi.org/10.1136/gutjnl-2013-306155 PMID: 24848265
- Padmanabhan P, Dixit NM. Mathematical model of viral kinetics in vitro estimates the number of E2-CD81 complexes necessary for hepatitis C virus entry. PLoS Comput Biol. 2011; 7: e1002307. https:// doi.org/10.1371/journal.pcbi.1002307 PMID: 22174670
- Brandenberg OF, Magnus C, Rusert P, Günthard HF, Regoes RR, Trkola A. Predicting HIV-1 transmission and antibody neutralization efficacy in vivo from stoichiometric parameters. PLoS Pathog. 2017; 13: e1006313. https://doi.org/10.1371/journal.ppat.1006313 PMID: 28472201
- Mulampaka SN, Dixit NM. Estimating the threshold surface density of gp120-CCR5 complexes necessary for HIV-1 envelope-mediated cell-cell fusion. PLoS ONE. 2011; 6: e19941. https://doi.org/10.1371/ journal.pone.0019941 PMID: 21647388
- Magnus C, Rusert P, Bonhoeffer S, Trkola A, Regoes RR. Estimating the stoichiometry of human immunodeficiency virus entry. J Virol. 2009; 83: 1523–1531. https://doi.org/10.1128/JVI.01764-08 PMID: 19019953
- Kalemera M, Mincheva D, Grove J, Illingworth CJR. Building a mechanistic mathematical model of hepatitis C virus entry. PLoS Comput Biol. 2019; 15: e1006905. <u>https://doi.org/10.1371/journal.pcbi.</u> 1006905 PMID: 30883541
- 66. Padmanabhan P, Dixit NM. Inhibitors of hepatitis C virus entry may be potent ingredients of optimal drug combinations. Proc Natl Acad Sci U S A. 2017; 114: E4524–E4526. https://doi.org/10.1073/pnas. 1704531114 PMID: 28512229
- Yang X, Kurteva S, Ren X, Lee S, Sodroski J. Stoichiometry of envelope glycoprotein trimers in the entry of human immunodeficiency virus type 1. J Virol. 2005; 79: 12132–12147. <u>https://doi.org/10.1128/</u> JVI.79.19.12132-12147.2005 PMID: 16160141
- Brandenberg OF, Magnus C, Regoes RR, Trkola A. The HIV-1 entry process: a stoichiometric view. Trends Microbiol. 2015; 23: 763–774. https://doi.org/10.1016/j.tim.2015.09.003 PMID: 26541228
- Boianelli A, Nguyen VK, Ebensen T, Schulze K, Wilk E, Sharma N, et al. Modeling influenza virus infection: a roadmap for influenza research. Viruses. 2015; 7: 5274–5304. <u>https://doi.org/10.3390/v7102875</u> PMID: 26473911
- Melville K, Rodriguez T, Dobrovolny HM. Investigating different mechanisms of action in combination therapy for influenza. Front Pharmacol. 2018; 9. https://doi.org/10.3389/fphar.2018.01207 PMID: 30405419
- Wong VC, Fong LE, Adams NM, Xue Q, Dey SS, Miller-Jensen K. Quantitative evaluation and optimization of co-drugging to improve anti-HIV latency therapy. Cell Mol Bioeng. 2014; 7: 320–333. <u>https://doi.org/10.1007/s12195-014-0336-9 PMID: 26191086</u>
- 72. Iwata-Yoshikawa N, Okamura T, Shimizu Y, Hasegawa H, Takeda M, Nagata N. TMPRSS2 contributes to virus spread and immunopathology in the airways of murine models after coronavirus infection. J Virol. 2019; 93: e01815–18. https://doi.org/10.1128/JVI.01815-18 PMID: 30626688
- Zhou Y, Vedantham P, Lu K, Agudelo J, Carrion R, Nunneley JW, et al. Protease inhibitors targeting coronavirus and filovirus entry. Antivir Res. 2015; 116: 76–84. https://doi.org/10.1016/j.antiviral.2015. 01.011 PMID: 25666761
- Shirato K, Kanou K, Kawase M, Matsuyama S. Clinical isolates of human coronavirus 229E bypass the endosome for cell entry. J Virol. 2017;91. https://doi.org/10.1128/JVI.01387-16 PMID: 27733646
- 75. Shirato K, Kawase M, Matsuyama S. Wild-type human coronaviruses prefer cell-surface TMPRSS2 to endosomal cathepsins for cell entry. Virology. 2018; 517: 9–15. https://doi.org/10.1016/j.virol.2017.11. 012 PMID: 29217279

- 76. Coutard B, Valle C, de Lamballerie X, Canard B, Seidah NG, Decroly E. The spike glycoprotein of the new coronavirus 2019-nCoV contains a furin-like cleavage site absent in CoV of the same clade. Antivir Res. 2020; 176: 104742. https://doi.org/10.1016/j.antiviral.2020.104742 PMID: 32057769
- Hoffmann M, Kleine-Weber H, Pöhlmann S. A multibasic cleavage site in the spike protein of SARS-CoV-2 is essential for infection of human lung cells. Mol Cell. 2020. <u>https://doi.org/10.1016/j.molcel.</u> 2020.04.022 PMID: 32362314
- 78. Hoffmann M, Hofmann-Winkler H, Smith JC, Krüger N, Sørensen LK, Søgaard OS, et al. Camostat mesylate inhibits SARS-CoV-2 activation by TMPRSS2-related proteases and its metabolite GBPA exerts antiviral activity. bioRxiv. https://doi.org/10.1101/2020.08.05.237651 PMID: 32793911
- 79. Shen L, Rabi SA, Sedaghat AR, Shan L, Lai J, Xing S, et al. A critical subset model provides a conceptual basis for the high antiviral activity of major HIV drugs. Sci Transl Med. 2011; 3: 91ra63–91ra63. https://doi.org/10.1126/scitranslmed.3002304 PMID: 21753122
- Lokugamage KG, Hage A, Schindewolf C, Rajsbaum R, Menachery VD. SARS-CoV-2 is sensitive to type I interferon pretreatment. bioRxiv. https://doi.org/10.1101/2020.03.07.982264 PMID: 32511335
- Beauchemin CAA, McSharry JJ, Drusano GL, Nguyen JT, Went GT, Ribeiro RM, et al. Modeling amantadine treatment of influenza A virus in vitro. J Theor Biol. 2008; 254: 439–451. https://doi.org/10.1016/j. jtbi.2008.05.031 PMID: 18653201
- Jiang Y, van der Welle JE, Rubingh O, van Eikenhorst G, Bakker WAM, Thomassen YE. Kinetic model for adherent vero cell growth and poliovirus production in batch bioreactors. Process Biochem. 2019; 81: 156–164. https://doi.org/10.1016/j.procbio.2019.03.010 PMID: 31217725
- Ursache RV, Thomassen YE, van Eikenhorst G, Verheijen PJT, Bakker WAM. Mathematical model of adherent vero cell growth and poliovirus production in animal component free medium. Bioprocess Biosyst Eng. 2015; 38: 543–555. https://doi.org/10.1007/s00449-014-1294-2 PMID: 25294335
- 84. Padmanabhan P, Dixit NM. Viral kinetics suggests a reconciliation of the disparate observations of the modulation of Claudin-1 expression on cells exposed to hepatitis C virus. PLoS ONE. 2012; 7: e36107. https://doi.org/10.1371/journal.pone.0036107 PMID: 22545157
- Marino S, Hogue IB, Ray CJ, Kirschner DE. A methodology for performing global uncertainty and sensitivity analysis in systems biology. J Theor Biol. 254: 178–196. https://doi.org/10.1016/j.jtbi.2008.04.011 PMID: 18572196